Author’s response to reviews

Title: Determination of the optimal inspiratory pressure providing adequate ventilation while minimizing gastric insufflation using real-time ultrasonography in Chinese children: a prospective, randomized, double-blind study

Authors:

Xiaowei Qian (qxwzhejiangedu@163.com)
Qiong Hu (huqiong_1221@qq.com)
Hang Zhao (zhaohang17777@163.com)
Bo Meng (mengbo1206@163.com)
Yang Nan (nay0000@163.com)
Hong Cao (Caohongwz@163.com)
Qingquan Lian (lianqingquanmz@163.com)
Jun Li (lijunwzmu@126.com)

Version: 5 Date: 01 Aug 2017

Author’s response to reviews:

Dear Prof. Spadaro

We appreciate editor and reviewers very much for their constructive comments and suggestions on our manuscript entitled “Determination of the optimal inspiratory pressure providing adequate ventilation while minimizing gastric insufflation using real-time ultrasonography in Chinese children: a prospective, randomized, double-blind study”. (ID: BANE-D-17-00041R4).

We have studied reviewer’s comments carefully and have made revision. Attached please find the revised version, which we would like to submit for your kind consideration. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Jun Li
List of Responses

Vincenzo Russotto (Reviewer 1): Authors attempted to improve the quality of the flowchart (figure 3). However the label of the allocation should be modified (they wrote allocated to P8 for the first three groups instead of allocated to P8, allocated to P10 and allocated to P12).

Answer: We have modified the flowchart. Thanks!